Author's response to reviews

Title: Fluvoxamine for fatigue in primary biliary cirrhosis and primary sclerosing cholangitis: a randomised controlled trial [ISRCTN88246634]

Authors:

Pieter CJ ter Borg (pterborg@zonnet.nl)
Erik van Os (e.vanos@erasmusmc.nl)
Walter W van den Broek (w.vandenbroek@erasmusmc.nl)
Bettina E Hansen (b.hansen@erasmusmc.nl)
Henk R van Buuren (h.vanbuuren@erasmusmc.nl)

Version: 4 Date: 12 July 2004

Author’s response to reviews:

Since the reviewers suggested no further changes, no such changes have been made to the manuscript. The following formatting changes have been made according to the editor’s requests:
1. The tables have been placed with their legend
2. The figures have been cropped
3. Figure 2 has been uploaded in a different format
4. The ISRCTN has been included in the title as requested.